EO-1 Weekly Status Week of June 29 – July 5, 2007

Day of Year 180 - 186
Mission Day 2410 - 2416
Earth Observing-One (EO-1) General:

There were 121 Data Collection Events (DCEs) scheduled this week.  
EO-1 Spacecraft Subsystems:

All spacecraft subsystems functioned nominally with the exception of the Command & Data Handling (C&DH) telemetry playback anomaly reported previously.
Instruments
All instruments operated nominally this week.  Decontamination cycles were performed for ALI and Hyperion.  HSI began a deicing on 07/05/2007 at 23:55z (07-186).  The deicing ends next week on 07/06/2007 at 14:45z (07-187).  ALI’s outgassing cycle is performed in conjunction with the HSI deicing.
[image: image1]Technology Activities:
In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

Another teleconference was held between the Wildfire Research and Applications Program (WRAP) and the Advanced Information System Technology (AIST) Sensor Web collaborators on Thursday 05 July.  The AIST team is slated to participate in the upcoming OGC OWS-5 testbed demonstration that is to start the end of July.
Initial testing of the EO-1 fire sensor web workflow began during the week.  Figure 1 depicts the portion of the EO-1 fire sensor web workflow being tested.  EO-1 was tasked to take 3 images and perform the workflow up to automatic creation of the classified fire pixels.  George Mason University is configuring their Web Coverage Service- Transformation (WCS-T) interface next week.  Figure 2 depicts the total top level fire sensor web workflow for this summer.
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Figure 1  Portion of automated fire sensor web being tested
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Figure 2  Top level fire sensor web demo for summer 2007

Autonomous Sciencecraft Experiment (ASE)

ASE controlled EO-1 all week except for the period of 07-181/23:15z to 07-182/09:00z, during which a nominal and a modified lunar calibration were performed using an ATS load.  Release 4 of the ASE onboard software was delivered to the MOC on 5 July.  Uplink of Release 4 is to begin Friday, 6 July.

Ground System
Hardware Status

	Hardware
	Status
	Comments

	Primary R/T System
	
	Green
	

	Backup R/T System
	
	Green
	FOT has upgraded ASIST to 9.7.e on the backup R/T system.  Testing is currently underway.  Telemetry and commanding will be tested utilizing the flight software lab and Flatsat.

	bMOC at DataLynx
	
	Green
	

	Playback System
	
	Green
	

	Mission Planning
	
	Yellow
	VMware license issues are waiting to be resolved.  Once FOT can utilize VMware on the primary mission planning machine axle, parallel operations and testing will begin.

	Flight Dynamics
	
	Green
	MathWorks have been contacted regarding license transfers to the new machines to install Matlab.  FOT will be receiving a new version of FreeFlyer (6.0) along with the license files to use with the new hardware within the next week.  AI group will be holding a users seminar at GSFC on July 25th.  FOT will attend this seminar.

	DTAS Server
	
	Green
	

	ANS
	
	Green
	

	Flight Software Lab
	
	Green
	

	Admin System
	
	Green
	


Operations

The EO-1 FOT has implemented the requested changes to the lunar calibration sequence.  The AC instrument has been removed from the sequence and the calibration has been reduced to 4 scans.  The time saved will be used to increase settling times between slews to improve image quality.  A test of this new sequence occurred on day 07-182.  A nominal lunar calibration was planned at 02:59z, with the modified sequence being planned to occur at 06:16z.  The nominal calibration was conducted successfully.  The modified calibration was conducted, but did not record the second post-image calibration for HSI or its lamp calibration.  It is believed the increased length of the sequence caused a switch from science collection mode to standard operations mode before the end of calibration termination activities.  The FOT is currently investigating this matter further before planning to conduct any other modified lunar calibrations.

Systems Administration

No change from last report.
New Discrepancy Reports (DR)

There are no new requests this week.
New Enhancement Requests (ER)
There are no new requests this week  

New Operations Database Change Requests (DBCR)

There are no new requests this week.
Ground and Space Network

EOS Data and Operations System (EDOS), GSFC, MD, USA

There are no major problems to report.
McMurdo Ground Station (MGS), McMurdo, Antarctica

There are no major problems to report.
Wallops Ground Station (WGS), Wallops Island, VA, USA

WGS is scheduled to be down for maintenance until 17 July 2007.
DataLynx Ground Station (PF1), Poker Flat, AK, USA

There are no major problems to report.
Alaska Ground Station (AGS), Poker Flat, AK, USA 

Due to an antenna brake failure, the S-Band support on day 180 at 02:25z was not conducted; no images were lost since this was an S-Band only support.

Svalbard Ground Station (SGS), Longyearbyen, Norway
There are no major problems to report
Tracking and Data Relay Satellite System (TDRSS), White Sands, NM, USA

There are no major problems to report.
USGS Multi-Satellite Ground Station (LGS), Sioux Falls, SD, USA
LGS is now being scheduled only for contingencies.
Tasmanian Earth Resources Satellite Station, a.k.a. Hobart (HGS), Hobart, Australia

Australian Centre for Remote Sensing reports that TERSS has been having some technical issues.  The antenna will be down until further notice.
Cordoba Ground Station (CGS), Cordoba, Argentina

No passes were scheduled this week. 

Upcoming Events:

· ALI internal calibration (Type 1) to be conducted on July 9, 2007

· WGS scheduled to be down until July 17, 2007

Imagery Status:

Scenes and Engineering Cals planned for week of June 29 – July 5, 2007                          121
Total scenes and engineering calibrations planned for entire mission – approximately 34,947
Total Scenes:  ALI scenes in the Level 0 archive            31,652 (as of July 5, 2007)

                        Hyperion scenes in the Level 0 archive    31,399
Publications and Presentations Status (as of 07-05-07):

404 publications 

284 external presentations

51 articles and press releases
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